Author's response to reviews

Title: Using diffusion of innovation theory to understand the factors impacting patients' acceptance and usage of consumer e-health innovations: a longitudinal case study in a primary care clinic

Authors:

Xiaojun Zhang (xjz787@uowmail.edu.au)
Ping Yu (Ping@uow.edu.au)
Ir Ton A. M. Spil (a.a.m.spil@utwente.nl)
Jun Yan (jyan@uow.edu.au)

Version: 2 Date: 19 May 2014

Author's response to reviews:

Tables have been uploaded after the reference section.
Figures have been removed from the manuscript and uploaded to separate single files.